Author’s response to reviews

Title: A comparison between drug-eluting stent implantation and drug-coated balloon angioplasty in patients with left main bifurcation in-stent restenotic lesions

Authors:

Hyungdon Kook (kookhyungdon@gmail.com)

Hyung Joon Joo (drjoohj@gmail.com)

Jae Hyoung Park (jhpark3992@naver.com)

Soon Jun Hong (psyche94@hanmail.net)

Cheol Woong Yu (ycw717@naver.com)

Do-Sun Lim (dslmd@kumc.or.kr)

Version: 2 Date: 03 Feb 2020

Author’s response to reviews:

Ref: BCAR-D-19-01048R1
Answers to the editor’s comments
We thank the editor for thoughtful and constructive comments on our work. Issues were raised by the editor, and we have addressed these issues point-by-point and enclosed our comments. We have used blue color font to indicate the revised portions of our manuscript for the editor. We hope that the editor would be satisfied with our responses and find the revised manuscript suitable for publication in BMC Cardiovascular Disorders.

Responses to Editor (William Spenceley Jones):
Comment 1.
In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Reply to Comment 1:
Thank you for your important comment. As the editor pointed out, we specified the role of the funding body in this manuscript.
Kept this comment in mind, we modified the Funding section and revised the manuscript accordingly.

Page 15, Line 343-344
Korea University (K1719491). The funding institution had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Comment 2.
We notice that some authors are missing from the authors' contributions section. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:
http://www.biomedcentral.com/submissions/editorial-policies#authorship
Reply to Comment 2:
Thank you for your very important comments. As the editor pointed out, we specified each author’s individual contribution to this manuscript.
Kept this comment in mind, we modified the Author’s contributions section and revised the manuscript accordingly.

Page 15-16, Line 347-350
JHP contributed to data collection and analysis of data. SJH contributed to conception and design of the study. CWY contributed to data collection and interpretation of data. D-SL contributed to clinical trial design and data management, final approval for the manuscript.

Comment 3.
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Reply to Comment 3:
Thank you for your kind comments. As directed by the editor, we uploaded a clean version of the manuscript and all relevant tables/figures/additional files. We also reconfirmed that all figures, tables and additional/supplementary files are properly cited within the text.